Author's response to reviews

Title: Implications of a RAD54L polymorphism (2290C/T) in human meningiomas as a risk factor and/or a genetic marker

Authors:
Paola E Leone (PLEONE@puce.edu.ec)
Marta Mendiola (mmendiola@iib.uam.es)
Javier Alonso (fjaviera@iib.uam.es)
Cesar Paz-y-Mino (CPAZYMINO@puce.edu.ec)
Dr Angel Pestana (apestana@iib.uam.es)

Version: 2 Date: 26 Feb 2003

To the BMC Cancer editor

Please find enclosed the new version of the manuscript entitled “Implications of a RAD54L polymorphism in human meningiomas as a risk factor and/or a genetic marker”, that I submit for publication in BMC Cancer. We have included all the discretionary revisions proposed by the reviewer. In particular:
- The number of tumour samples is 70 and the typing error in Abstracts has been corrected.
- The epidemiological data of meningioma patients is shown in a new table (Table 1) and a discussion is provided in the first paragraph of Results
- Emphasis on the differences in C/T polymorphism between Ecuadorian and Spanish, together with an explanation of the discrepancies is given in the second paragraph of Results.
- The quality of figure 1 with SSCP results has been upgraded.

Angel Pestana
apestana@iib.uam.es

Files and formats supplied:

Manuscript: bmcRAD.doc
Figures Fig 01. ppt
Fig 02. ppt
Fig 03. ppt